Author's response to reviews

Title: Antimicrobial and Cytotoxicity Properties of the Crude Extracts and Fractions of Premna resinosa (Hochst.) Schauer (Compositae): Kenyan Traditional Medicinal Plant

Authors:

Sospeter N Njeru (hicogn@gmail.com)
Meshack A Obonyo (obonyom@gmail.com)
Samwel O Nyambati (sonsarigo07@gmail.com)
Silas M Ngari (msngari@yahoo.com)

Version: 2
Date: 20 December 2014

Author's response to reviews:

Dear Editor,

I have removed all sections break from the article
I have also inserted page numbers and line numbers to the article.

Regards
Sospeter